RESEARCH

Open Access

Body mass index and health-related quality of life of outpatients with knee osteoarthritis: evidence from a cross-sectional study

Xueshan Sun¹, Xuemei Zhen^{2,3}, Kaijie Liu⁸, Wenqianzi Yang⁴, Shuyan Gu⁵, Xu Zhang⁶ and Hengjin Dong^{4,7*}

Abstract

Purpose To estimate the health-related quality of life (HRQoL) of knee osteoarthritis outpatients in China overall and by body mass index (BMI), and to explore the factors associated with knee osteoarthritis.

Method This was a cross-sectional study in which outpatients with knee osteoarthritis were recruited from three tertiary hospitals in China from June 2020 to May 2021. The EuroQoL five-dimensional instrument was used to measure HRQoL. Descriptive analysis, one-way analysis of variance, and a Tobit regression model were performed.

Results One thousand and eight questionnaires were distributed, and nine hundred and fifty-two participants were included. The mean age was 61.71 years, and 61.03% of the participants were female. The mean health utility was 0.751 (95% CI 0.737, 0.765). The health utility varied significantly according to BMI: underweight = 0.627 (95% CI 0.536, 0.719), normal = 0.764 (95% CI 0.744, 0.784), overweight = 0.755 (95% CI 0.733, 0.776), and obese = 0.728 (95%CI 0.681, 0.776). Additionally, outpatients with knee osteoarthritis who were younger, had received treatment in the past six months, and had a disease duration shorter than 2 years had significantly higher HRQoL scores.

Conclusion The HRQoL of knee osteoarthritis outpatients was considerably impaired and was significantly associated with BMI. These findings emphasize the importance of weight examination and management in outpatients with knee osteoarthritis for healthcare providers. Supplementary data were provided for public health-related policies and health economics studies, contributing to the development of effective management strategies and increasing the breadth of knowledge of HRQoL in outpatients with knee osteoarthritis.

Plain English Summary

There were 595 million people living with osteoarthritis globally in 2020, equal to 7.60% of the global population, with an increase of 132.20% in total cases since 1990. The knee was the most prevalent joint with a prevalence of 21.51% among people aged 40 years and older in China, while the weight has been proven to be an important modifiable risk factor for knee osteoarthritis. Considering the increasing cases of overweight/obesity in China and the great disease burden on individuals and the healthcare system, we estimated the effect of knee osteoarthritis on health and associated factors, health-related quality of life (HRQoL) was measured using health utility and the EQ-VAS scores. A total of 952 participants were included, the average health utility was 0.751, and the average

*Correspondence: Hengjin Dong donghj@zju.edu.cn

Full list of author information is available at the end of the article



© The Author(s) 2025. **Open Access** This article is licensed under a Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 International License, which permits any non-commercial use, sharing, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if you modified the licensed material. You do not have permission under this licence to share adapted material derived from this article or parts of it. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by-nc-nd/4.0/.



EQ-VAS score was 70.37. Participants with a normal BMI were found to have the highest HRQoL, indicating the importance of managing weight in patients with knee osteoarthritis to improve HRQoL. Moreover, age, history of treatment and disease duration have been proven to be associated factors, which provides evidence for public health policymakers and health economic professionals. The findings of the present study increase the breadth of knowledge of HRQoL in individuals with knee osteoarthritis.

Keywords Health-related quality of life, Knee osteoarthritis, Utility, BMI, EQ-5D-5L

Introduction

Osteoarthritis is one of the leading causes of disability and pain worldwide, characterized by joint pain and loss of mobility, imposing a great disease burden on patients and the healthcare system [1]. The World Health Organization (WHO) has designated the period from 2021 to 2030 as the decade of healthy ageing, highlighting the need to improve quality of life and address chronic diseases, such as osteoarthritis [2]. According to a report based on the Global Burden of Disease Study (GBD), there were 595 million people living with osteoarthritis globally in 2020, equal to 7.6% of the global population, with an increase of 132.2% in total cases since 1990 [2]. Osteoarthritis can affect every synovial joint, and the knee is the most frequently affected joint. Among people aged 40 years and older, the prevalence of knee osteoarthritis in China was 21.51% [3]. Osteoarthritis has been responsible for a major source of health expenditure, in the USA, osteoarthritis caused an estimated US\$80 billion in 2016, and in Hong Kong, osteoarthritis caused an estimated more than \$400 million in medical spending [2].

Osteoarthritis not only increases medical expenditure of the healthcare system, but also seriously impairs patients' health-related quality of life (HRQoL) [4-6]. HRQoL is a multidimensional measure of subjective feelings towards and experiences with health, involving physical, mental and functional health [7]. HRQoL is a widely recognized measure to estimate the effect of diseases especially for chronic diseases on people [8]. Osteoarthritis is responsible for great joint pain, increasing disability and progressive cartilage degeneration in people with osteoarthritis, and significantly affecting HRQoL. The measurement of HRQoL for people with osteoarthritis can provide health care professionals with a better understanding of the disease burden of osteoarthritis [9]. Additionally, the utility estimates are essential input parameters for measuring quality-adjusted life years (QALYs) in economic evaluations, which may vary depending on multiple factors such as population and disease severity [8].

Osteoarthritis places significant disease burdens on both the HRQoL of patients and the increasing medical expenditures in healthcare systems. However, currently, no treatment strategies can cure osteoarthritis completely, modifiable risk factors are important for delaying the onset of osteoarthritis and slowing the progression of osteoarthritis. It has been proven that several possible factors may be associated with the onset and progression of osteoarthritis: age, sex, BMI, genetic factors, behavioral influences, ethnicity, and the environment [6, 10–12]. Among these factors, BMI has been considered to be the main established modifiable risk factor [2, 13, 14]; moreover, approximately half of adults and onefifth of children are forecast to be overweight or obese in China according to the Chinese criteria, making China the country with the highest number of people with overweight or obesity in the world [15]. Considering the proven association between BMI and osteoarthritis and the increasing cases of overweight and obese in China, reliable estimation of HROoL for osteoarthritis patients with different BMI levels is necessary to fully understand the disease burden.

Previous studies have estimated the HRQoL of knee osteoarthritis patients using disease-specific instruments, such as the Western Ontario and MacMaster questionnaire (WOMAC) [16-18], the Lysholm knee score [19, 20], the Oxford Knee Score and the Activity and Participation Questionnaire [21, 22], and the Osteoarthritis of Knee and Hip Quality of Life questionnaire (OQKHQOL) [23]. Although these disease-specific instruments have been able to measure changes in disease status for patients with knee osteoarthritis, they are not able to provide the results of health utility scores, which are urgently needed by health-related policies and health economic models. Therefore, in this study, generic instrument was used to report health utility scores for patients with knee osteoarthritis. Meanwhile, although the association has been proven and BMI is supposed to be a modifiable risk factor for osteoarthritis [2, 13, 14], few studies have estimated the extent to which BMI affects the HRQoL of people with knee osteoarthritis [24, 25]. Therefore, the present study aims to estimate the health utility of outpatients with knee osteoarthritis in China by using the EQ-5D-5 L instrument and to explore the factors associated with the HRQoL of outpatients with knee osteoarthritis, with a special emphasis on BMI. The present study can serve as a valuable supplement to previous studies and provide important data for health economics professionals and public health policymakers.

Methods

Study design and participants

This was a cross-sectional study, which was reviewed and approved by the Medical Faculty Ethics Committee of Zhejiang University. Throughout all phases of the research, the confidentiality and the anonymity of the collected data were ensured. Participants' names, contact information and other privacy-related information were not involved in the questionnaire or the face-to-face investigation, and virtual unique codes were assigned to participants during data analysis.

Outpatients with knee osteoarthritis were recruited from three tertiary hospitals in Zhejiang Province from June 2020 to May 2021 (Fig.S1). The eligibility criteria were as follows: (1) outpatients aged 40 years and older; (2) outpatients who were diagnosed with knee osteoarthritis by a physician according to the criteria outlined in the "Osteoarthritis Diagnosis and Treatment Guideline (2018)" established by the Joint Surgery Group of Chinese Orthopaedics Association (Table S2) [26]; (3) outpatients who were not currently pregnant; and (4) outpatients who were conscious, had no cognitive impairment, psychological problems or mental illness, and could express themselves clearly. Outpatients were excluded if they had (1) other serious diseases, such as cardiovascular disease, liver and kidney function impairment, immune deficiency or blood disease, or (2) other bone diseases or injuries that may affect their normal joint function.

Instruments and variable description

The survey was conducted by using a self-designed questionnaire, which was designed according to previous studies and expert consultation [7–9]. The questionnaire was composed of three parts: sociodemographic variables, clinical variables, and a five-level EuroQol fivedimensional instrument (EQ-5D-5 L).

The sociodemographic variables included sex, age, education level, marital status, and healthcare insurance scheme. Age was categorized into three groups: <55, 55-64, and ≥ 65 years. Education level was categorized into four levels: no formal education, primary school, middle school, and high school and above. Marital status was classified as single (not married or not living in a partnership) or not single (married or living in a partnership). The insurance schemes were divided into four types: no insurance, basic medical insurance for urban employees (UEBMI), basic medical insurance for urban residents (URBMI) and commercial medical insurance.

The clinical variables included BMI, duration of knee osteoarthritis, and history of treatment in the past six months. The cut-off values for BMI in this study were based on the "weight for adults" criterion established by the National Health Commission of China [27]. The participants were classified into four categories based on their BMI values: underweight (BMI < 18.5 kg/m²), normal weight ($18.5 \le BMI < 24.0 \text{ kg/m}^2$), overweight ($24.0 \le BMI < 28.0 \text{ kg/m}^2$) and obese (BMI ≥ 28.0 kg/m²) [27]. The duration of knee osteoarthritis was divided into three categories: <2 years, 2–5 years, and >5 years. A history of treatment was a binary variable that referred to whether the outpatient had received any type of treatment in the past six months, including conservative/surgical/medication treatment.

The EQ-5D-5 L instrument was applied in the present study, which is widely used to measure HRQoL, and a Chinese version was used [28-30]. The EQ-5D-5 L comprises two components: the health state descriptive system and the visual analogue scale (EQ-VAS) of 100 mm. In the health state system, a 5-point response scale was applied in the descriptive system, where 1 refers to no problem, 2 refers to a slight problem, 3 refers to a moderate problem, 4 refers to a severe problem, and 5 refers to an extreme problem [30]. The responses to the description were converted into utility index scores through a Chinese general population-based EQ-5D-5 L value set based on the time trade-off (TTO) technique [31]. In the EQ-VAS, participants marked their health on a scale ranging from 0 to 100, where 0 refers to the worst self-reported health state and 100 refers to the best selfreported health state.

Data collection

A face-to-face investigation was conducted. First, the investigators contacted the clinician to review the medical records to confirm the diagnosis of knee osteoarthritis, and then further checked whether the outpatients met the inclusion criteria. Second, if an outpatient met all the inclusion criteria, informed consent was provided to confirm whether the outpatient agreed to participate. Third, for outpatients who met the inclusion criteria and agreed to participate, a face-to-face interview was conducted. Before the investigation, two investigators from the research group were trained based on uniform standards (one was a PhD candidate, and the other was a master's student from our research group at Zhejiang University) to ensure that they fully understood the purpose of the study and the context of the questionnaire. During the investigation, if problems were found, investigators would try to ask the participants again to ensure data integrity and availability. After the investigation, the data were double entered to ensure the accuracy of the database.

Statistical analysis

There were three steps to analyze the data in the present study. First, descriptive statistics were adopted to report the characteristics and HRQoL of knee OA outpatients, and subgroup analysis was conducted to further explore the data based on demographic and clinical characteristics. The frequencies with percentages were calculated for categorical variables, and the means and 95% confidence intervals (CIs) were calculated for continuous variables. Second, one-way analysis of variance (ANOVA) was applied to compare the health utility and EQ-VAS score according to participant characteristics. Third, a Tobit regression model was adopted to explore the factors that were significantly associated with health utility and the EQ-VAS score. The Tobit model was chosen because the distributions of both health utility and the EQ-VAS score were skewed [31]. Independent variables were chosen based on previous studies, including sex, age, marital status, educational level, insurance scheme, BMI, history of treatment, and duration of knee osteoarthritis [6, 10-12]. A significance level of 0.05 was used. The data were double entered into EpiData 3.1 and analyzed using SPSS 26.0 and Stata/SE 15.1.

Results

Participant characteristics

A total of 1008 questionnaires were distributed, with the exclusion of 42 questionnaires lacking key information and 14 questionnaires having logical problems. As a result, 952 knee osteoarthritis outpatients were included in the present study (Fig.S1). A total of 41 participants were underweight, 467 had a normal BMI, 350 were overweight and 94 were obese. The mean age was 61.71 years (95% CI 61.08, 62.34), with a male proportion of 38.97%. More than half of the participants (50.10%) had an education level of middle school, high school or above. A total of 86.66% of the participants were not single, and more than half of the participants (81.62%) reported a duration of knee OA shorter than 5 years (Table 1).

Health-related quality of life

The most prevalent problem reported by participants with knee osteoarthritis was pain/discomfort (96.11%), followed by impaired activity level (62.50%). The impaired self-care level was the least reported (33.40%). When the participants were divided into different BMI groups, the most serious problem was still pain/discomfort: 92.68% of underweight participants, 96.79% of normal weight participants, 96.29% of overweight participants, and 93.62% of obese participants. The second most serious problems among underweight, normal weight, overweight, and obese participants groups were anxiety/ depression (73.17%), usual activity (60.17%), usual activity (63.14%), and usual activity (68.09%), respectively (Fig. 1; Table S1).

The mean health utility of outpatients with knee osteoarthritis was 0.751 (95% CI 0.737, 0.765). The health utility varied significantly between participants with different BMI (P = 0.002). The health utility was 0.627 (95% CI 0.536, 0.719) for underweight participants, 0.764 (95% CI 0.744, 0.784) for normal weight participants, 0.755 (95% CI 0.733, 0.776) for overweight participants, and 0.728 (95% CI 0.681, 0.776) for obese participants (Fig. 2; Table 1).

The mean EQ-VAS score for knee osteoarthritis outpatients was 70.37 (95% CI 69.64, 71.09). The EQ-VAS scores were statistically significantly different between participants with different BMI (P=0.013). The score was 67.71 (95% CI 63.28, 72.14) for underweight participants, 71.15 (95% CI 70.18, 72.12) for normal weight participants, 70.43 (95% CI 69.30, 71.57) for overweight participants, and 67.38 (95% CI 64.25,70.52) for obese participants (Table 1).

Factors associated with health-related quality of life

Univariate analysis revealed significant associations between HRQoL (both health utility and EQ-VAS scores) and age, educational level, insurance scheme, BMI, and duration of disease; additionally, associations between health utility and history of treatment and between the EQ-VAS score and marital status were also found. Associations between HRQoL scores and sex were not significant. Participants who were younger, had a higher education level, were covered by UEBMI insurance, or had a normal BMI had greater health utility and EQ-VAS scores (Table 1).

However, the multivariate regression showed that education level and insurance scheme were not significant factors associated with health utility or the EQ-VAS score after adjusting for other variables. The strongest determinant of health utility was BMI. Compared with underweight participants, those with a normal BMI, overweight status, or obesity had greater health utility, while participants with a normal BMI had the highest health utility. In terms of the EQ-VAS score, compared with underweight participants, normal weight and overweight participants had higher EQ-VAS scores, while obese participants had lower EQ-VAS scores. Additionally, compared with participants aged younger than 55 years, those aged older than 65 years had lower EQ-5D utility. Compared with participants who did not receive treatment in the past six years, those who received treatment had greater health utility. Compared with those with a duration shorter than 2 years, those whose duration was longer than 5 years had lower health utility. Compared with participants younger than 55 years, those older than 55 years had lower EQ-VAS scores, compared with those with a duration shorter than 2 years, those whose duration was longer than 5 years had lower EQ-VAS scores (Table 2).

Table 1 Characteristics of knee osteoarthritis outpatients, and health utility and EQ-VAS score stratified by their characteristics

Variables	N (%)	Health utility			EQ-VAS score		
		Mean	95%CI	Р	Mean	95%CI	Р
Total	952(100)	0.751	[0.737,0.765]		70.37	[69.64,71.09]	
Sex				0.043			0.068
Male	371(38.97)	0.769	[0.748,0.791]		71.21	[69.99,72.44]	
Female	581(61.03)	0.739	[0.721,0.758]		69.83	[68.92,70.73]	
Age (years)				< 0.001			< 0.001
< 55	232(24.37)	0.810	[0.786,0.833]		74.04	[72.69,75.40]	
55–64	315(33.09)	0.766	[0.743,0.788]		70.59	[69.32,71.86]	
≥65	405(42.54)	0.706	[0.682,0.730]		68.09	[66.97,69.20]	
Education level				< 0.001			< 0.001
No formal education	158(16.60)	0.710	[0.670,0.751]		68.33	[66.53,70.14]	
Primary school	317(33.30)	0.712	[0.684,0.739]		69.00	[67.79,70.22]	
Middle school	274(28.78)	0.776	[0.754,0.798]		70.40	[68.99,71.81]	
High school and above	203(21.32)	0.811	[0.787,0.834]		74.04	[72.55,75.53]	
Marital status ^a				0.402			0.011
Single	127(13.34)	0.736	[0.696,0.775]		67.95	[65.57,70.34]	
Not single	825(86.66)	0.753	[0.738,0.768]		70.74	[69.98,71.49]	
Insurance				< 0.001			< 0.001
No insurance	45(4.73)	0.728	[0.655,0.801]		70.09	[66.11,74.06]	
UEBMI	266(27.94)	0.810	[0.787,0.832]		73.05	[72.01,74.09]	
URBMI	569(59.77)	0.726	[0.706,0.745]		69.47	[68.47,70.47]	
Commercial medical insurance	72(7.56)	0.749	[0.703,0.794]		67.71	[64.91,70.51]	
History of treatment				0.019			0.615
Untreated	135(14.18)	0.710	[0.668,0.753]		69.91	[68.03,71.78]	
Treated	817(85.82)	0.758	[0.743,0.772]		70.44	[69.65,71.23]	
BMI ^b				0.002			0.013
Underweight	41(4.31)	0.627	[0.536,0.719]		67.71	[63.28,72.14]	
Normal	467(49.05)	0.764	[0.744,0.784]		71.15	[70.18,72.12]	
Overweight	350(36.76)	0.755	[0.733,0.776]		70.43	[69.30,71.57]	
Obese	94(9.87)	0.728	[0.681,0.776]		67.38	[64.25,70.52]	
Duration (years) ^c				< 0.001			< 0.001
<2	296(31.09)	0.767	[0.743,0.790]		71.78	[70.58,72.98]	
2–5	481(50.53)	0.771	[0.752,0.789]		70.82	[69.86,71.79]	
>5	175(18.38)	0.670	[0.630,0.711]		66.73	[64.65,68.80]	

BMI body mass index, CI confidence interval; ^aSingle refers to not married or not living in a partnership; not single refers to married or living in a partnership; ^bBMI category was defined based on "weight for adults" criterion established by the National Health Commission of China, the cut-off values for BMI: underweight (BMI < 18.5 kg/m²), normal weight (18.5 \leq BMI < 24.0 kg/m²), overweight (24.0 \leq BMI < 28.0 kg/m²), and obese (BMI \geq 28.0 kg/m²); ^cDuration refers to the time elapsed from the initial diagnosis of knee osteoarthritis to the time of investigation

Discussion

This cross-sectional study estimated the HRQoL and its association with BMI and other potential factors using the EQ-5D-5 L. Furthermore, HRQoL is vital for measuring the impact of knee osteoarthritis on health, and it was represented by both health utility and the EQ-VAS scores in the present study. This study was the first to estimate not only the health utility and EQ-VAS scores for knee osteoarthritis outpatients but also the distribution of HRQoL among individuals with different BMI groups in China. The findings revealed an average health utility of 0.751 for all included knee osteoarthritis outpatients, with participants having a normal BMI exhibiting the highest health utility. Similarly, the average EQ-VAS score was 70.37, with participants having a normal BMI also exhibiting the highest EQ-VAS score.

Outpatients with knee osteoarthritis suffer from considerable impairment in HRQoL. On the one hand, when compared with the normal population, the health utility of the general Chinese population was estimated using the EQ-5D-5 L instrument, with an estimated health utility of 0.927 [32]. This value was much higher than that reported in the present study, emphasizing the critical importance of the management and prevention of knee osteoarthritis. On the other hand, the estimated health utility in our study was higher than that reported in the UK [33]. The National Institute for Health and Care Excellence (NICE) reported health utility values of 0.688, 0.701, and 0.723 for patients with osteoarthritis



Fig. 1 Distribution of the EQ-5D problems reporting among knee osteoarthritis outpatients



Fig. 2 EQ-5D utility scores of knee osteoarthritis outpatients by weight

who did not take any medications, took paracetamol or took NSAIDs/COX-2 inhibitors, respectively [33]. The cultural difference between China and the UK may contribute non-negligibly to this disparity. All items in the EQ-5D-5 L were subjective, encompassing participants' attitude towards pain and self-care. Compared to participants in the UK, Chinese people have maintained the world's most enduring and successful continuous culture over the past four millennia, with primary beliefs in Taoism, Confucianism, and Buddhism influencing them for thousands of years [34, 35]. For example, within Buddhism beliefs, pain and suffering are viewed as a power, unwanted yet existent, that arises from obstacles in one's past life; within Confucianism, pain is considered an essential element of life. Therefore, it is more likely that Chinese patients will experience pain but also demonstrate resilience in enduring it [35]. Apart from cultural differences, medications were not distinguished in the present study, thus there is a possibility that medication may have influenced participants' attitudes and potentially overestimated people's self-reported health status in the present study.

BMI was indicated to be associated with the HRQoL of participants with knee osteoarthritis in the present study,

Table 2	Association between knee osteoarthritis outpatients' characteristics and EQ-5D uti	ility, EQ-VAS	score using a	Tobit regression
model				

Variables	Health utility			EQ-VAS s	EQ-VAS score		
	Coef	95%Cl	Р	Coef	95%Cl	P	
Sex (ref = male)							
Female	-0.017	[-0.044,0.011]	0.245	-0.853	[-2.304,0.598]	0.249	
Age, years (ref = < 55)							
55–64	-0.025	[-0.062,0.012]	0.189	-2.400	[-4.322,-0.478]	0.014	
≥65	-0.066	[-0.104,-0.029]	0.001	-4.235	[-6.196,-2.274]	< 0.001	
Education level (ref=no formal education)							
Primary school	-0.016	[-0.057,0.024]	0.433	-0.217	[-2.331,1.897]	0.841	
Middle school	0.024	[-0.020,0.068]	0.293	-0.326	[-2.624,1.972]	0.781	
High school and above	0.038	[-0.011,0.087]	0.133	2.443	[-0.115,5.001]	0.061	
Marital status ^a (ref=single)							
Not single	-0.010	[-0.051,0.030]	0.611	1.437	[-0.650,3.524]	0.177	
Insurance (ref = no insurance)							
UEBMI	0.038	[-0.031,0.106]	0.279	0.587	[-2.976,4.150]	0.747	
URBMI	-0.010	[-0.074,0.054]	0.762	-0.751	[-4.085,2.584]	0.659	
Commercial medical insurance	0.006	[-0.072,0.085]	0.873	-2.917	[-7.004,1.169]	0.162	
History of treatment (ref=untreated)							
Treated	0.053	[0.015,0.092]	0.007	0.967	[-1.044,2.978]	0.346	
BMI ^b (ref=underweight)							
Normal	0.123	[0.056,0.191]	< 0.001	2.736	[-0.777,6.248]	0.127	
Overweight	0.112	[0.043,0.180]	0.001	1.785	[-1.788,5.357]	0.327	
Obese	0.104	[0.026,0.181]	0.009	-0.417	[-4.451,3.616]	0.839	
Duration (ref = < 2)							
2–5	0.009	[-0.021,0.040]	0.550	-0.578	[-2.177,1.020]	0.478	
>5	-0.075	[-0.115,-0.035]	< 0.001	-3.920	[-5.996,-1.843]	< 0.001	

Ref reference group, BMI body mass index, CI confidence interval, Coef coefficient; ^aSingle refers to not married or not living in a partnership; not single refers to married or living in a partnership

^bBMI category was defined based on the "weight for adults" criterion established by the National Health Commission of China, the cut-off values for BMI: underweight (BMI < 18.5 kg/m²), normal weight (18.5 \leq BMI < 24.0 kg/m²), overweight (24.0 \leq BMI < 28.0 kg/m²), and obese (BMI \geq 28.0 kg/m²); ^cDuration refers to the time elapsed from the initial diagnosis of knee osteoarthritis to the time of investigation

and participants with a normal weight had the highest HRQoL scores. The finding that overweight and obese participants had worse HRQoL scores is consistent with previous studies [36-39], which highlights the importance of weight loss for overweight/obese patients with knee osteoarthritis to improve their HROoL in clinic. It was indicated individuals with an elevated BMI were at a greater risk of experiencing knee pain compared to those with a normal BMI [39]. Body weight represents a potentially modifiable risk factor for the development of knee osteoarthritis, while for obese patients with osteoarthritis, weight loss can help them alleviate symptoms when present [38]. The association between weight and HRQoL among osteoarthritis patients deserves increased attention, and overweight and obese patients with knee osteoarthritis are advised to lose weight to improve their HRQoL, as it has been revealed that 10% weight loss can significantly improve physical HRQoL for overweight or obese patients with knee osteoarthritis [39]. Meanwhile, when compared with participants with a normal BMI, those with an underweight BMI also experienced significantly poorer HRQoL. In the present study, in addition to pain, anxiety/depression was found to be the second most serious problem for underweight participants. However, this was not the case for other participants with different BMI groups, which may explain the poorer HRQoL observed among underweight individuals. Xie's study also found a worse mental HRQoL among underweight participants [7]. They measured the HRQoL of patients with knee osteoarthritis using the 12-item Short Form (SF-12), and underweight participants had worse mean mental component summary (MCS) scores [7]. Additionally, previous studies have reported a U-shaped relationship between anxiety/depression and BMI, with increased levels of anxiety/depression among both underweight and overweight/obese individuals [40, 41]. This finding may partly explain the poorer HRQoL observed among underweight and overweight/obese participants in this study. Meanwhile, the results discussed above focus on statistical differences, and it is also essential to further explore the findings of this study from the perspective of clinical differences. Although there are no validated Minimal Clinically Important Differences (MCIDs) for EQ-5D-5 L health utility values and

EQ-5D-VAS especially for patients with knee osteoarthritis, and significant variations in MCID cut-offs exist due to differences in models and populations [42, 43], there are other relevant studies reporting MCID values of health utility. For example, a 10% improvement was used as an MCID for patients undergoing total knee arthroplasty [44], while another study adopted a threshold of 0.085 for patient undergoing primary knee arthroplasty [45]. Additionally, a threshold of 0.1 was used for patients with rheumatoid arthritis [46]. Therefore, based on the range of MCID thresholds reported in previous studies, the health utility values of patients with a normal BMI, an overweight BMI, and an obese BMI are clinically significantly higher than those of patients with an underweight BMI.

In addition to BMI, potential factors of age, history of treatment and duration of disease were significantly associated with HRQoL for participants with knee osteoarthritis in the present study. A poorer HRQoL was observed among older participants, which is consistent with most studies [7, 9, 47, 48]. Osteoarthritis is a degenerative disease, and it is undeniable that age is recognized as the main influencing factor of osteoarthritis. Symptoms typically onset in middle-aged and elderly individuals, and with increasing age, worsening clinical symptoms may have an impact on the HRQoL of elderly patients. However, Wang's study drew different conclusions on the impact of age on HRQoL, they found that older patients had better HRQoL [49]. They hypothesized that elderly patients who have been suffering from osteoarthritis for a long time may have adapted to the disease, therefore the impact of osteoarthritis on their HRQoL was not as significant as that on younger patients who have had the disease for a shorter duration in their study [49]. History of treatment also significantly affected the HRQoL of participants. Compared with participants who did not receive any treatment in the past six months, participants who received any type of treatment had better HRQoL in the present study, and this finding was partly consistent with two recent studies [7, 50]. Xie's study demonstrated that participants who received any type of treatment had worse physical component summary (PCS) scores, despite having higher MCS [7]. Maybe the fact that all participants included were outpatients with knee osteoarthritis can partly account for this difference. Most of the participants had early-stage knee osteoarthritis and treatments may help reduce the level of pain/anxiety, which in turn benefits patients' HRQoL in the present study. Schepman's study also found that non-pharmacological treatment can predict a higher quality of life [50]. Longer duration of disease negatively affects patients' HRQoL in the present study, which is consistent with the characteristics and progression of knee osteoarthritis. Knee osteoarthritis is a chronic degenerative disease, and longer duration is usually accompanied by more severe symptoms or comorbidities, which may result in worse HRQoL [7].

With the increasing global burden of knee osteoarthritis, effective management of the disease needs to be implemented. In view of the findings of the present study, we suppose the following: First, we highlight the importance of weight examination and management in clinical settings. Patients with a healthy BMI had the highest HRQoL scores, representing a gap and opportunity to improve HRQoL by incorporating dietary guidelines, nutritional education and anxiety relief [51]. Specifically, most studies advised weight loss to improve patients' HRQoL, while weight management for participants with an underweight BMI received less attention. Health intervention strategies should be taken for unhealthy weight patients, which include patients with overweight, obese, and underweight BMI. Second, our study found that age and duration of disease were negatively associated with HRQoL. Since the two factors are non-modifiable, we strongly emphasize that policymakers in the field of public health can prioritize policies aimed at preventing and delaying osteoarthritis progression, in order to alleviate the negative impact of age and duration of disease on HRQoL. Third, the results indicate that participants with a history of treatment in the past six months had better HRQoL, which was partly consistent with existing studies. Therefore, we agree with the treatment principles recommended in the Chinese Osteoarthritis Diagnosis and Treatment Guideline (2018) [26], which suggest that physicians should consider factors such as age, sex, weight, and risk factors, and provide individualized treatment for the patients. Ultimately, since the overall health utility and utility by characteristics were reported for Chinese outpatients with knee osteoarthritis, the results can serve as evidence for future health intervention and cost-effectiveness studies in the field of osteoarthritis management.

Several limitations in the present study should be mentioned. First, the study was conducted in three tertiary hospitals, which may not be truly representative of all patients with knee osteoarthritis in China, thus limiting the generalization of the results, however, the present study can add additional evidence to the existing HRQoL studies for patients with knee osteoarthritis. Second, as this study was a cross-sectional study, only associations could be estimated, and the causes of the difference in HRQoL could not be inferred from the present findings. Therefore, a longitudinal study that includes all of the hospital care systems in China is needed in the future to reach a solid conclusion on the associations found in the present study. Third, almost half of the participants were without formal education or had attended only primary school, and possible misunderstandings may have

existed, which could have biased the responses. Training courses were conducted for the investigators to explain the items to the participants in an easily understandable and accurate way.

Conclusion

The HRQoL of outpatients with knee osteoarthritis was considerably impaired according to our findings, which implied that improved management is important to prevent the rapid progression of knee osteoarthritis. BMI is a modifiable risk factor for knee osteoarthritis, and our findings also revealed a higher HRQoL in outpatients with a normal BMI. The management of weight in overweight/obese outpatients with knee osteoarthritis is vital. We estimated the health utility and EQ-VAS score for outpatients with knee osteoarthritis overall and by BMI class, which could be used in future public health-related policies and economic studies to increase the breadth of knowledge of HRQoL in outpatients with knee osteoarthritis. The factors associated with knee osteoarthritis found in the present study can also guide efforts to improve the HRQoL of outpatients with knee osteoarthritis in the clinic.

Supplementary Information

The online version contains supplementary material available at https://doi.or g/10.1186/s12891-025-08432-2.

Supplementary Material 1

Acknowledgements

We would like to thank all study participants for their participation. We would like to thank Li Chen, Meiling Zhang, Risheng Zhang, Weifang Zheng, Zhen Chen, Jiahao Xu in the investigated hospitals in Zhejiang Province, who made significant efforts in the period of investigation.

Author contributions

Author contributions X.S., K.L. and H.D. conducted the investigation; X.S., X.Z., S.G. and H.D. discussed and wrote the methodology; H.D. supervised the entire investigation and writing; X.S and X.Z. wrote the original draft; W.Y., X.Z. and H.D. reviewed and edited the manuscript. All authors read and approved the final manuscript.

Funding

The authors declare that no funds, grants, or other support were received during the preparation of this manuscript.

Data availability

All of the principal data were included in the results in the present study. Additional materials with further details can be obtained from the corresponding author.

Declarations

Ethics approval

The authors are accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved. This study was approved by the institutional review board of Zhejiang University School of Public Health (20180923).

Consent to participate

Informed consent was obtained from all participants included in the present study.

Consent to publish

This manuscript didn't contain any individual person's data in any form.

Competing interests

The authors declare no competing interests.

Author details

¹School of Public Policy and Administration, Xi'an Jiaotong University, Xi'an, Shannxi, China

²School of Health Care Management, Shandong University, 44 Wenhuaxi Rd, Jinan, Shandong, China

³NHC Key Laboratory of Health Economics and Policy Research,

Shandong University, 44 Wenhuaxi Rd, Jinan, Shandong, China

⁴Center for Health Policy Studies, School of Public Health, Zhejiang

University School of Medicine, Hangzhou, Zhejiang, China

⁵Center for Health Policy and Management Studies, School of

Government, Nanjing University, Nanjing, Jiangsu, China

⁶Department of Sport Science, College of Education, Zhejiang University, Hangzhou, Zhejiang, China

⁷The Fourth Affiliated Hospital Zhejiang University School of Medicine, Yiwu, Zhejiang, China

⁸Taizhou Hospital of Zhejiang, Linhai, Zhejiang, China

Received: 24 October 2023 / Accepted: 14 February 2025 Published online: 05 March 2025

References

- Boer CG, Hatzikotoulas K, Southam L, Stefansdottir L, Zhang Y, Coutinho de Almeida R, Wu TT, Zheng J, Hartley A, Teder-Laving M, Skogholt AH, Terao C, Zengini E, Alexiadis G, Barysenka A, Bjornsdottir G, Gabrielsen ME, Gilly A, Ingvarsson T, Johnsen MB, Jonsson H, Kloppenburg M, Luetge A, Lund SH, Magi R, Mangino M, Nelissen R, Shivakumar M, Steinberg J, Takuwa H, Thomas LF, Tuerlings M, Ogen Consortium arc, Hunt All-In Pain, Argo Consortium, Center Regeneron Genetics, Babis GC, Cheung JPY, Kang JH, Kraft P, Lietman SA, Samartzis D, Slagboom PE, Stefansson K, Thorsteinsdottir U, Tobias JH, Uitterlinden AG, Winsvold B, Zwart JA, Davey Smith G, Sham PC, Thorleifsson G, Gaunt TR, Morris AP, Valdes AM, Tsezou A, Cheah KSE, Ikegawa S, Hveem K, Esko T, Wilkinson JM, Meulenbelt I, Lee MTM, J. B. J., van Meurs U, Styrkarsdottir, Zeggini E. Deciphering Osteoarthritis Genetics across 826,690 Individuals from 9 Populations. *Cell* 184, no. 18 (2021): 4784–818 e17.
- Collaborators O. Global, Regional, and National Burden of Osteoarthritis, 1990–2020 and Projections to 2050: A Systematic Analysis for the Global Burden of Disease Study 2021. *Lancet Rheumatol* 5, no. 9 (2023): e508–e22.
- Sun X, Zhen X, Hu X, Li Y, Gu S, Gu Y, Dong H. Osteoarthritis in the Middleaged and Elderly in China: prevalence and influencing factors. Int J Environ Res Public Health. 2019;16:23.
- Xie F, Li SC, Goeree R, Tarride JE, O'Reilly D, Lo NN, Yeo SJ, Yang KY, Thumboo J. Validation of Chinese Western Ontario and Mcmaster universities osteoarthritis Index (Womac) in patients scheduled for total knee replacement. Qual Life Res. 2008;17(4):595–601.
- Grassel S, Muschter D. Recent Advances in the Treatment of Osteoarthritis. F1000Res 9 (2020).
- Miyamoto Y, Shi D, Nakajima M, Ozaki K, Sudo A, Kotani A, Uchida A, Tanaka T, Fukui N, Tsunoda T, Takahashi A, Nakamura Y, Jiang Q, Ikegawa S. Common variants in Dvwa on chromosome 3p24.3 are Associated with susceptibility to knee osteoarthritis. Nat Genet. 2008;40(8):994–8.
- Xie Y, Yu Y, Wang JX, Yang X, Zhao F, Ma JQ, Chen ZY, Liang FR, Zhao L, Cai DJ. Yang. Health-Related Quality of Life and its influencing factors in Chinese with knee osteoarthritis. Qual Life Res. 2020;29(9):2395–402.
- Gu S, Wang X, Shi L, Sun Q, Hu X, Gu Y, Sun X, Dong H. Health-Related Quality of Life of Type 2 diabetes patients hospitalized for a diabetes-related complication. Qual Life Res. 2020;29(10):2695–704.
- Vitaloni M, Botto-van Bemden A, Sciortino Contreras RM, Scotton D, Bibas M, Quintero M, Monfort J, Carne X, de Abajo F, Oswald E, Cabot MR, Matucci M, du Souich P, Moller I, Eakin G, Verges J. Global management of patients

with knee osteoarthritis begins with quality of Life Assessment: a systematic review. BMC Musculoskelet Disord. 2019;20(1):493.

- Allen KD, Thoma LM, Golightly YM. Epidemiol Osteoarthr Osteoarthr Cartil. 2022;30(2):184–95.
- 11. Slemenda CW. The epidemiology of Osteoarthritis of the knee. Curr Opin Rheumatol. 1992;4(4):546–51.
- Thabit Z, Al-Qazaz H. Health-Related Quality of Life among patients with osteoarthritis: a cross-sectional study. Georgian Med News. no. 2023;334:65–70.
- 13. Hu Y, Chen X, Wang S, Jing Y, Su J. Subchondral bone microenvironment in Osteoarthritis and Pain. Bone Res. 2021;9(1):20.
- 14. Kulkarni K, Karssiens T, Kumar V, Pandit H. Obes Osteoarthr Maturitas. 2016;89:22–8.
- Wang Y, Zhao L, Gao L, Pan A, Xue H. Health Policy and Public Health Implications of Obesity in China. Lancet Diabetes Endocrinol. 2021;9(7):446–61.
- Xu L, Wang C, Zhang C, Feng X, Tong W. Cross-cultural adaption and validation of simplified Chinese version of the lower extremity function scale in patients with knee osteoarthritis. Clin Rheumatol. 2020;39(10):3041–48.
- Shen ZD, Yu HM, Wang JT, Shi GY, Sun Y. [Modified western Ontario and Mcmaster University Osteoarthritis Index Scale used in patients with knee osteoarthritis]. Zhonghua Yi Xue Za Zhi. 2019;99(7):537–41.
- Cheung RT, Ngai SP, Ho KK. Chinese adaptation and validation of the knee Injury and Osteoarthritis Outcome score (Koos) in patients with knee osteoarthritis. Rheumatol Int. 2016;36(10):1449–54.
- Xie F, Thumboo J, Lo NN, Yeo SJ, Yang KY, Yeo W, Chong HC, Fong KY, Li SC. Cross-cultural adaptation and validation of Singapore English and Chinese versions of the Lequesne Algofunctional Index of knee in asians with knee osteoarthritis in Singapore. Osteoarthritis Cartilage. 2007;15(1):19–26.
- Wang W, Liu L, Chang X, Jia ZY, Zhao JZ, Xu WD. Cross-cultural translation of the Lysholm knee score in Chinese and its validation in patients with Anterior Cruciate Ligament Injury. BMC Musculoskelet Disord. 2016;17(1):436.
- Xie F, Li SC, Lo NN, Yeo SJ, Yang KY, Yeo W, Chong HC, Fong KY, Thumboo J. Cross-cultural adaptation and validation of Singapore English and Chinese versions of the Oxford knee score (Oks) in knee osteoarthritis patients undergoing total knee replacement. Osteoarthritis Cartilage. 2007;15(9):1019–24.
- Chen C, Wang W, Wu H, Gao A, Qiu Y, Weng W, Price A. Cross-cultural translation and validation of the Chinese Oxford knee score and the activity and participation questionnaire. J Orthop Surg (Hong Kong). 2020;28(2):2309499020910668.
- Wang W, He CR, Zheng W, Li J, Xu WD. Development of a Valid simplified Chinese Version of the osteoarthritis of knee and hip quality of life (Oakhqol) in patients with knee or hip osteoarthritis. J Eval Clin Pract. 2016;22(1):53–61.
- 24. Riddle DL, Stratford PW. Body weight changes and corresponding changes in Pain and function in persons with symptomatic knee osteoarthritis: a Cohort Study. Arthritis Care Res (Hoboken). 2013;65(1):15–22.
- 25. Belibagli MC, Gokmen MY. Five-year impact of weight loss on knee Pain and Quality of Life in obese patients. Med Sci Monit. 2025;31:e946550.
- Association, Joint Surgery Group of Chinese Orthopaedic. [Osteoarthritis diagnosis and treatment Guideline (2018 version)]. Chin J Orthop. 2018;38(12):11.
- 27. National Health Commission of the People's Republic of China. Criteria weight adults. 2013. Accessed 06 Feb 2025.
- Beaudart C, Biver E, Bruyere O, Cooper C, Al-Daghri N, Reginster JY, Rizzoli R. Quality of Life Assessment in musculo-skeletal health. Aging Clin Exp Res. 2018;30(5):413–18.
- Luo N, Li M, Liu GG, Lloyd A, de Charro F, Herdman M. Developing the Chinese Version of the New 5-Level Eq-5d descriptive system: the Response Scaling Approach. Qual Life Res. 2013;22(4):885–90.
- Office EQ. Eq-5d, Helping the World Make Better Health Decisions. (accessed 02/06).
- 31. Luo N, Liu G, Li M, Guan H, Jin X. Rand-Hendriksen. Estimating an Eq-5d-5l value set for China. Value Health. 2017;20(4):662–69.
- Zhou J, Xu L, Pan J, Wang M, Zhou P, Wang W, Lu S, Zhu W. A comparative study of Chinese Medicine Quality of Life Assessment Scale (Cq-11d) and Eq-5d-5l and Sf-6d scales based on Chinese Population. Qual Life Res. 2024;33(1):113–22.
- Excellence, National Institute for Health and Care. Osteoarthritis: Care and Management in Adults. In Osteoarthritis: Care and Management in Adults. London, 2014.

- 34. Hsu CY, O'Connor M, Lee S. Understandings of death and dying for people of Chinese origin. Death Stud. 2009;33(2):153–74.
- Chen LM, Miaskowski C, Dodd M, Pantilat S. Concepts within the Chinese Culture that influence the Cancer Pain Experience. Cancer Nurs. 2008;31(2):103–8.
- Rogers MW, Wilder FV. The Association of Bmi and knee Pain among persons with radiographic knee osteoarthritis: a cross-sectional study. BMC Musculoskelet Disord. 2008;9:163.
- Abbate LM, Jordan JM. Weight Change Osteoarthr Osteoarthr Cartil. 2012;20(4):268–70.
- 38. Felson DT. Weight and osteoarthritis. Am J Clin Nutr. 1996;63(3):S430–32.
- Messier SP, Resnik AE, Beavers DP, Mihalko SL, Miller GD, Nicklas BJ, deVita P, Hunter DJ, Lyles MF, Eckstein F, Guermazi A. Loeser. Intentional weight loss in overweight and obese patients with knee osteoarthritis: is more better? Arthritis Care Res (Hoboken). 2018;70(11):1569–75.
- Jung SJ, Woo HT, Cho S, Park K, Jeong S, Lee YJ, Kang D, Shin A. Association between body size, Weight Change and Depression: systematic review and Meta-analysis. Br J Psychiatry. 2017;211(1):14–21.
- Lee JH, Park SK, Ryoo JH, Oh CM, Choi JM, McIntyre RS, Mansur RB, Kim H, Hales S. Jung. U-Shaped relationship between Depression and Body Mass Index in the Korean adults. Eur Psychiatry. 2017;45:72–80.
- Paulsen A, Djuv A, Dalen I. Clinical cut-offs for hip- and knee arthroplasty outcome - minimal clinically important improvement (Mcii) and patient acceptable symptom state (pass) of patient-reported outcome measures (Prom). Qual Life Res. 2025. https://doi.org/10.1007/s11136-025-03896-0. Epub ahead of print.
- 43. Yang M, Kondo T, Talebi A, Jhund PS, Docherty KF, Claggett BL, Vaduganathan M, Bachus E, Hernandez AF, Lam CSP, Inzucchi SE, Martinez FA, de Boer RA, Kosiborod MN, Desai AS, Kober L, Ponikowski P, Sabatine MS, Solomon SD. V. McMurray. Dapagliflozin and quality of life measured using the Euroqol 5-Dimension questionnaire in patients with heart failure with reduced and mildly Reduced/Preserved ejection fraction. Eur J Heart Fail. 2024;26(7):1524–38.
- 44. Golinelli D, Polidoro F, Rosa S, Puzzo A, Guerra G, Raimondi S, Chiaravalloti A, Sisti V, Sanmarchi F, Bravi F, Grilli R, Pia Fantini M, Belluati A. Evaluating the Impact of Robotic-Assisted Total Knee Arthroplasty on quality of care through patient-reported outcome measures in a third-level hospital in Italy: a prospective cohort study. Knee. 2025;52:32–42.
- Yapp LZ, Scott CEH, Howie CR, MacDonald DJ, Simpson A. Clement. Meaningful values of the Eq-5d-3l in patients undergoing primary knee arthroplasty. Bone Joint Res. 2022;11(9):619–28.
- Lahiri M, Cheung PPM, Dhanasekaran P, Wong SR, Yap A, Tan DSH, Chong SH, Tan CH, Santosa A, Phan P. Evaluation of a Multidisciplinary Care Model to Improve Quality of Life in Rheumatoid Arthritis: a Randomised Controlled Trial. Qual Life Res. 2022;31(6):1749–59.
- Ettinger WH, Davis MA, Neuhaus JM. Mallon. Long-term physical functioning in persons with knee osteoarthritis from Nhanes. I: effects of Comorbid Medical conditions. J Clin Epidemiol. 1994;47(7):809–15.
- Cuzdan Coskun N, Ay S, Evcik FD, Oztuna D. Adiponectin: is it a biomarker for assessing the Disease severity in knee osteoarthritis patients? Int J Rheum Dis. 2017;20(12):1942–49.
- Jingxuan W, Yue X, Zhihao X, Yong YU, Zhiyu C, Chunxia Y. [Investigation on the quality of life of knee osteoarthritis patients in Sichuan Area and its influencing factors]. J Prev Med Inform. 2021;37(3):6.
- Schepman P, Robinson R, Blakeman KH, Wilhelm S, Beck C, Hallberg S, Liseth-Hansen J, De Geer A, Rolfson O, Arendt-Nielsen L. Factors influencing quality of life in patients with osteoarthritis: analyses from the biscuits Study. Scand J Pain. 2023;23(1):139–48.
- Gomes-Neto M, Araujo AD, Junqueira ID, Oliveira D, Brasileiro A, Arcanjo FL. Comparative study of functional capacity and quality of life among obese and non-obese Elderly people with knee osteoarthritis. Rev Bras Reumatol Engl Ed. 2016;56(2):126–30.

Publisher's note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.